Mon Apr 2 17:02: 06 GMI 2001
Year: 01 Doy: 092
Qobserver: yasukawa
WEATHER COMMENT: Mon Apr 2 17:02:31 GMVI 2001
Cool, in cirrus and altocumul us overcast, brisk south wind.
GONG COMMENT: Mon Apr 2 17:03: 06 GMIr 2001
Performed weekly and nonthly preventive mai nt enance before com ng
up to observatory since we are in heavy overcast and there is no
rush to start up M.SO i nstrunentation.
Mon Apr 2 20:31:53 GMr 2001
IV

Mon Apr 2 22:08:41 GMIr 2001
Year: 01 Doy: 092
ohserver: yasukawa

Mon Apr 2 22:10:12 GMI 2001 CH P Start Patrol
Mon Apr 2 22:12:26 GMI 2001 MKI V Start Patrol
Mon Apr 2 22:12:52 GMr 2001 CH P End Patr ol

Mon Apr 2 22:27:09 GMVI 2001 Pl CS Start Patrol

**M.SO PROBLEM *: NMon Apr 2 22:35:34 GMI 2001

Did the network changeover today. All systens are now on ml so.ucar. edu
now. All systens except nene have been checked out--albeit there is no
sunshine here today. David is currently troubl eshooting nene to see why
the diskwite is not happening.

Mon Apr 2 23:04:37 GVII 2001 Pl CS Start Patrol

Mon Apr 2 23:13:06 GMVI 2001 Pl CS End Patr ol

**Pl CS PROBLEM*: Mon Apr 2 23:20:20 GVr 2001

PI CS was not running the canmera task. It ran when David started up a

Canera task remptely. The plan is for me to reboot nene to spawn the
camera task and we will check it out tonorrow.
Mon Apr 2 23:22:00 GVI 2001

IV

22 12.rawnk4 22 15.rawrk4



